Author's response to reviews

Title: Correlation of antiangiogenic, antioxidant and cytotoxic activities of some Sudanese medicinal plants with phenolic and flavonoid contents

Authors:

Loiy Elsir A Hassan (loiy.ahmed23@gmail.com)
Mohamed Khadeer Ahamed Basheer (khadeer.nc@gmail.com)
Aman S Abdul Majid (amanshah75@yahoo.com)
Husin Baharetha (husinmso@yahoo.com)
Nahdzatul S Muslim (syima.muslim@gmail.com)
Zeyad D Nassar (zeiadnassar@yahoo.com)
Amin Malik S Abdul Majid (aminmalikshah@gmail.com)

Version: 4 Date: 26 September 2014

Author's response to reviews:

Response to all the comments from reviewers is given in the attached letter.